Author's response to reviews

Title:Current parental tobacco use is associated with higher Body Mass Index in children and adolescents: an international cross-sectional study

Authors:

Irene E Braithwaite (irene.braithwaite@mrinz.ac.nz)
Alistair W Stewart (aw.stewart@auckland.ac.nz)
Robert J Hancox (bob.hancox@otago.ac.nz)
Richard Beasley (richard.beasley@mrinz.ac.nz)
Rinki Murphy (r.murphy@auckland.ac.nz)
Edwin A Mitchell (e.mitchell@auckland.ac.nz)

Version:2 Date:26 March 2015

Author's response to reviews:

Many thanks for your email dated 26 March requesting I make changes to the submitted manuscript prior to the review process.

I apologise for any inconvenience this may have caused.

In response to your requests:
1. Authors qualifications have now been removed from the title page
1.1. The email addresses for all authors have been added to the title page, I was unsure if there was a specific format for this, and have added each email after each author's name
2.0 "Introduction" heading has now been changed to "Background"
3.0 Authors' contributions. This has been reworded to align better with your suggested wording, and has been moved from after the abstract to line 312 on Page 16. I hope this meets your requirements.

Thank you again and apologies for any inconvenience caused

Regards
Irene Braithwaite
Corresponding author.